

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Benati, Luca; Surico, Paolo

## Working Paper VAR analysis and the Great Moderation

ECB Working Paper, No. 866

**Provided in Cooperation with:** European Central Bank (ECB)

*Suggested Citation:* Benati, Luca; Surico, Paolo (2008) : VAR analysis and the Great Moderation, ECB Working Paper, No. 866, European Central Bank (ECB), Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/153300

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# WORKING PAPER SERIES NO 866 / FEBRUARY 2008

VAR ANALYSIS AND THE GREAT MODERATION

by Luca Benati and Paolo Surico









In 2008 all ECB publications feature a motif taken from the €10 banknote.



# WORKING PAPER SERIES

**NO 866 / FEBRUARY 2008** 

# VAR ANALYSIS AND THE GREAT MODERATION'

Luca Benati<sup>2</sup> and Paolo Surico<sup>3</sup>

This paper can be downloaded without charge from http://www.ecb.europa.eu or from the Social Science Research Network electronic library at http://ssrn.com/abstract\_id=1090280.

 We wish to thank F. Canova, T. Cogley, G. Corsetti, M. Gertler, M. Giannoni, C. Julliard, T. Lubik, B. Mojon, G. Primiceri, J. Rubio-Ramirez, F. Ravenna, F. Schorfheide, A. Scott, I. Wolden Bache, and T. Zha for helpful discussions, seminar participants at the 2007 NBER Summer Institute, the Federal Reserve Board, the Federal Reserve Banks of New York and Richmond, the American Economic Association 2008 meetings, the London School of Economics, Universitat Autonoma de Barcelona, the Bank of Spain, the 2006 meetings of the Royal Economic Society and the Society for Computational Economics, the 5th Workshop on Macroeconomics Dynamics held at the Bank of Italy and the 2nd Oslo Workshop on Monetary Policy for comments. The views expressed in this paper are those of the authors, and do not necessarily reflect those of the European Central Bank or the Bank of England, 2 Monetary Policy Strategy Division, European Central Bank, Kaiserstrasse 29, D-60311 Frankfurt am Main, Germany; e-mail: Luca.Benati@ecb.int

3 Bank of England and University of Bari; Postal address: Monetary Policy Committee Unit, Bank of England, Threadneedle Street, London, EC2R 8AH, UK; e-mail: Paolo.Surico@bankofengland.gsi.gov.uk

#### © European Central Bank, 2008

**Address** Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postal address Postfach 16 03 19 60066 Frankfurt am Main, Germany

**Telephone** +49 69 1344 0

Website http://www.ecb.europa.eu

**Fax** +49 69 1344 6000

All rights reserved.

Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the author(s).

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb. europa.eu/pub/scientific/wps/date/html/ index.en.html

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

# CONTENTS

Ał	ostrac	t	4
No	on-tec	hnical summary	5
1	Intro	oduction	7
	1.1	Main results	7
	1.2	Explaining the results	8
2	Asse	essing VAR studies of the Great Moderation	9
	2.1	The model	10
	2.2	Bayesian estimation	13
3	Theo	pretical properties of the estimated	
	data-	-generation process	14
	3.1	representations of the model under	
		the two regimes	14
	3.2	The VARMA representations	16
	3.3	Impulse-response functions of	
		the model under the two regimes	17
4	Repl	icating the Great Moderation	17
	4.1	Volatility decreases in population	17
	4.2	Generating 'Great Inflations' and	
		'Great Moderations'	17
5	Can	structural VAR methods	
	unco	over the truth?	18
	5.1	I heoretical structural	19
	52	Impulse-response functions	20
6	Som	e criticisme of our analysis	20
0	and	our rebuttals	21
	6.1	Canova's criticism	21
	6.2	A further criticism	24
7	Con	clusions	25
Re	feren	ces	26
Α	Two	technical aspects of the	
11	Bave	esian estimation procedure	28
	A.1	Numerical maximisation of the	
		log posterior	28
	A.2	Calibrating the covariance	
		matrix scale factor	28
В	Wha	t is the additional state variable	
	unde	er indeterminacy?	29
	B.I	Reframing the issue in the correct way	29
	D.2	VAR representations of the DSGE model	30
Тс	blag o	nd figures	20
га	oles a	nu ngules	27
Εu	ropea	in Central Bank Working Paper Series	57

#### Abstract

Most analyses of the U.S. Great Moderation have been based on structural VAR methods, and have consistently pointed towards good luck as the main explanation for the greater macroeconomic stability of recent years. Based on an estimated New-Keynesian model in which the *only* source of change is the move from passive to active monetary policy, we show that VARs may misinterpret good policy for good luck.

First, the policy shift is *sufficient* to generate decreases in the theoretical innovation variances for all series, and decreases in the variances of inflation and the output gap, without any need of sunspot shocks. With sunspots, the estimated model exhibits decreases in both variances and innovation variances for all series. Second, policy counterfactuals based on the theoretical structural VAR representations of the model under the two regimes fail to capture the truth, whereas impulse-response functions to a monetary policy shock exhibit little change across regimes.

Since these results are in line with those found in the structural VARbased literature on the Great Moderation, our analysis suggests that existing VAR evidence is compatible with the 'good policy' explanation of the Great Moderation.

Keywords: Great Moderation; DSGE models; indeterminacy; vector autoregressions.

JEL codes: E38, E52.

## Non Technical Summary

Post-WWII U.S. macroeconomic history is usually divided into two distinct subperiods. The former period, which extends up to the end of the Volcker disinflation, is characterised by a significant extent of macroeconomic turbulence, with highly volatile inflation and output growth. The latter period, from the end of the Volcker disinflation up to the present day, is marked in contrast by significantly smaller volatilities for both inflation and output growth. These dramatic changes in the reduced-form properties of the U.S. economy over the last several decades characterise a phenomenon known as the 'Great Moderation'.

A vast empirical literature has investigated the source(s) of the Great Moderation in an attempt to disentangle the relative contributions of two main explanations: good policy and good luck. Based on (time-varying or Markov-switching) structural VAR methods, the good luck hypothesis has been advocated by a number of authors including Stock and Watson (2002), Primiceri (2005), Sims and Zha (2006), and Gambetti, Pappa, and Canova (2006). Based on estimated sticky-price DSGE models of the U.S. economy, both Lubik and Schorfheide (2004) and Boivin and Giannoni (2006) find, in contrast, support for the good policy explanation originally advocated by Clarida, Gali, and Gertler (2000), according to which a shift in the systematic component of monetary policy has been the driving force behind the recent, greater macroeconomic stability.

This paper tries to reconcile the two conflicting sets of results by asking whether *methodological* differences between the two approaches might account for the differences in their outcomes. In order to investigate the ability of structural VAR methods to correctly identify the sources of the Great Moderation, we use as data-generation process a New Keynesian model in which—in line with Clarida, Gali, and Gertler (2000) and Lubik and Schorfheide (2004)—the *only* sources of change are the move from passive to active monetary policy, and the presence of sunspots under indeterminacy. We estimate the model *via* Bayesian methods, and we explore the theoretical properties of the estimated structure.

Our main results may be summarised as follows.

First, the shift in the systematic component of monetary policy associated with the move from indeterminacy to determinacy is sufficient to generate, in population, (i) decreases in the innovation variances for all series, and (i) decreases in the variances of inflation and the output gap, as a simple implication of the Lucas (1976) critique, and without any need of sunspot shocks. With sunspot shocks, the estimated model exhibits decreases in both variances and innovation variances in population when moving from indeterminacy to determinacy, thus replicating the key features of the Great Moderation.

Second, policy counterfactuals based on the theoretical structural VAR representations of the model under the two regimes fail to capture the truth. In particular, substituting the VAR's structural monetary rule corresponding to the indeterminacy regime into the VAR for the determinacy regime causes a volatility *decrease*—rather than an increase—for two series out of three.

*Third*, impulse-response functions to a monetary policy shock exhibit little change across regimes.

Overall, our results suggests that existing VAR evidence is, in principle, uninformative on the issue of the role played by monetary policy in the Great Moderation, and is compatible with the notion that policy played a crucial role in fostering the greater macroeconomic stability of recent years.

We identify two key dimensions along which VAR analysis turns out to be misleading. First, in general, changes in the coefficients of the monetary policy rule of the DSGE model exert their impact on *both* the coefficients of the VAR representation of the model, *and* the elements of the VAR's covariance matrix of reduced-form innovations. Although this is a well-known implication of the Lucas (1976) critique, this point has generally been overlooked in the structural VAR-based empirical literature on the Great Moderation, which has routinely interpreted changes in the volatilities of the reduced-form innovations, accompanied by weak evidence of changes in the VAR's coefficients, as evidence against good policy, and in favor of good luck. As this paper shows, however, the dominant impact of a change in the systematic component of monetary policy may well turn out to be the one on the elements of the VAR's covariance matrix, with a comparatively milder effect on the VAR's coefficients. As a corollary, this logically implies that this kind of evidence does not allow, in principle, to discriminate among the good policy and good luck explanations, simply because, within our data generation process, they are essentially observationally equivalent.

Second, changes in the interest rate equation (i.e, the monetary policy rule) of a structural VAR bear no clear-cut relationship with changes in the parameters of the monetary policy rule in the underlying DSGE model. To put it differently, there appears to be a fundamental disconnect between what *is* structural within a DSGE model, and what is *defined* as structural based on the structural VAR representation implied by the very same DSGE model. Earlier contributions, on the other hand, have performed counterfactual simulations in structural VARs under the implicit *presumption* that switching the estimated coefficients of the interest rate equations in the structural VAR provides a reasonable approximation to the *authentic* policy counterfactual, i.e. the one you obtain by switching the parameters of the monetary policy rule in the underlying DSGE model. Again, as the present work shows, such a presumption is, in general, unjustified.

Finally, the present work contribues to the literature along another dimension, by identifying a crucial, and previously unnoticed difference between the determinacy and indeterminacy regimes for New Keynesian models. In particular, under indeterminacy the equivalent minimal state-space representation of the DSGE model possesses an additional state variable compared with the determinacy regime.

## 1 Introduction

Post-WWII U.S. macroeconomic history is usually divided into two distinct subperiods. The former period, which extends up to the end of the Volcker disinflation, is characterised by a significant extent of macroeconomic turbulence, with highly volatile inflation and output growth. The latter period, from the end of the Volcker disinflation up to the present day, is marked in contrast by significantly smaller volatilities for both inflation and output growth. These dramatic changes in the reduced-form properties of the U.S. economy over the last several decades characterise a phenomenon known as the 'Great Moderation'.<sup>1</sup>

A vast empirical literature has investigated the source(s) of the Great Moderation in an attempt to disentangle the relative contributions of two main explanations: good policy and good luck. Based on (time-varying or Markov-switching) structural VAR methods, the good luck hypothesis has been advocated by a number of authors including Stock and Watson (2002), Primiceri (2005), Sims and Zha (2006), and Gambetti, Pappa, and Canova (2006) (the disaggregated analysis of Mojon (2007), based on a Markov-switching structural VAR, finds however an important role for the unsystematic component of monetary policy in fostering the Great Moderation). Based on estimated sticky-price DSGE models of the U.S. economy, both Lubik and Schorfheide (2004) and Boivin and Giannoni (2006) find, in contrast, support for the good policy explanation originally advocated by Clarida, Gali, and Gertler (2000), according to which a shift in the systematic component of monetary policy has been the driving force behind the recent, greater macroeconomic stability.

This paper tries to reconcile the two conflicting sets of results by asking whether *methodological* differences between the two approaches might account for the differences in their outcomes. In order to investigate the ability of structural VAR methods to correctly identify the sources of the Great Moderation, we use as data-generation process a New Keynesian model in which—in line with Clarida, Gali, and Gertler (2000) and Lubik and Schorfheide (2004)—the *only* sources of change are the move from passive to active monetary policy,<sup>2</sup> and the presence of sunspots under indeterminacy. We estimate the model *via* Bayesian methods, and we explore the theoretical properties of the estimated structure.

### 1.1 Main results

Our main results may be summarised as follows.

• The shift in the systematic component of monetary policy associated with the

<sup>&</sup>lt;sup>1</sup>See in particular Kim and Nelson (1999) and McConnell and Perez-Quiros (2000).

 $<sup>^{2}</sup>$ As we abstract from the role of fiscal policy, the relationship between the monetary policy stance and equilibrium (in)determinacy in a simple New-Keynesian model is one-to-one, with a passive (active) rule associated with an indeterminate (determinate) equilibrium. As shown by Leeper (1991), in more complex settings this is not the case.

move from indeterminacy to determinacy is *sufficient* to generate, *in population*, (*i*) decreases in the innovation variances for all series, and (*ii*) decreases in the variances of inflation and the output gap, as a simple implication of the Lucas (1976) critique, and without any need of sunspot shocks. With sunspot shocks, the estimated model exhibits decreases in both variances and innovation variances in population when moving from indeterminacy to determinacy, thus replicating the key features of the Great Moderation.

- Policy counterfactuals based on the theoretical structural VAR representations of the model under the two regimes fail to capture the truth. In particular, substituting the VAR's structural monetary rule corresponding to the indeterminacy regime into the VAR for the determinacy regime causes a volatility *decrease*—rather than an increase—for two series out of three.
- Impulse-response functions to a monetary policy shock exhibit little change across regimes.

Overall, our results suggests that existing VAR evidence is, in principle, uninformative on the issue of the role played by monetary policy in the Great Moderation, and is compatible with the notion that policy played a crucial role in fostering the greater macroeconomic stability of recent years.

#### 1.2 Explaining the results

We identify two key dimensions along which VAR analysis turns out to be misleading.

First, in general, changes in the coefficients of the monetary policy rule of the DSGE model exert their impact on *both* the coefficients of the VAR representation of the model, *and* the elements of the VAR's covariance matrix of reduced-form innovations. Although this is a well-known implication of the Lucas (1976) critique, this point has generally been overlooked in the structural VAR-based empirical literature on the Great Moderation, which has routinely interpreted changes in the volatilities of the reduced-form innovations, accompanied by weak evidence of changes in the VAR's coefficients, as evidence against good policy, and in favor of good luck. As this paper shows, however, the dominant impact of a change in the systematic component of monetary policy may well turn out to be the one on the elements of the VAR's coefficients. As a corollary, this logically implies that this kind of evidence does not allow, in principle, to discriminate among the good policy and good luck explanations, simply because, within our data generation process, they are essentially observationally equivalent.

Second, changes in the interest rate equation (i.e, the monetary policy rule) of a structural VAR bear no clear-cut relationship with changes in the parameters of the monetary policy rule in the underlying DSGE model. To put it differently, there appears to be a fundamental disconnect between what *is* structural within a DSGE model, and what is *defined* as structural based on the structural VAR representation implied by the very same DSGE model. Earlier contributions, on the other hand, have performed counterfactual simulations in structural VARs under the implicit *presumption* that switching the estimated coefficients of the interest rate equations in the structural VAR provides a reasonable approximation to the *authentic* policy counterfactual, i.e. the one you obtain by switching the parameters of the monetary policy rule in the underlying DSGE model. Again, as the present work shows, such a presumption is, in general, unjustified.

Finally, the present work contribues to the literature along another dimension, by identifying a crucial, and previously unnoticed difference between the determinacy and indeterminacy regimes for New Keynesian models. In particular, under indeterminacy the equivalent minimal state-space representation of the DSGE model possesses an additional state variable compared with the determinacy regime.

The paper is organized as follows. Section 2 describes the standard New Keynesian model we use in the paper, and discusses details of both the Bayesian estimation procedure and the specific experiment we construct. Section 3 discusses key theoretical properties of the estimated data-generation process, focusing on the difference between the equivalent minimal state-space representations of the model under the two regimes, which imply that the model possesses a VAR representation under determinacy, and a VARMA one under inderminacy. Section 4 shows hos the estimated structure replicates key aspects of the Great Moderation in population. Section 5 shows how neither structural VAR-based policy counterfactuals, nor impulse-response functions to a monetary policy shock, point towards the authentic cause of changes in the data-generation process. Section 6 concludes.

## 2 Assessing VAR Studies of the Great Moderation

In order to assess the ability of structural VAR methods to correctly identify the causes of the Great Moderation, we consider the following experiment:

Suppose that the Great Moderation in the United States has been exclusively due to improved monetary policy, with a passive monetary policy regime in place before October 1979, and an active regime in place thereafter. Would structural VAR techniques be capable of uncovering the authentic causes of the changes in the data-generation process?

As we will see, the answer is 'No', with structural VAR methods clearly pointing towards 'good luck'—i.e., an exogenous reduction in the variance of the structural shocks—as the true underlying cause of the changes in the DGP, in spite of the fact that, by construction, *everything* is here driven by improved monetary policy.

#### 2.1 The model

The model we use in what follows is given by

$$R_t = \rho R_{t-1} + (1-\rho) [\phi_\pi \pi_t + \phi_y y_t] + \epsilon_{R,t}$$
(1)

$$\pi_t = \frac{\beta}{1+\alpha\beta} \pi_{t+1|t} + \frac{\alpha}{1+\alpha\beta} \pi_{t-1} + \kappa y_t + \epsilon_{\pi,t}$$
(2)

$$y_t = \gamma y_{t+1|t} + (1-\gamma)y_{t-1} - \sigma^{-1}(R_t - \pi_{t+1|t}) + \epsilon_{y,t}$$
(3)

where  $R_t$ ,  $\pi_t$  and  $y_t$  are the nominal interest rate, inflation, and the output gap, respectively;  $\alpha$  and  $\gamma$  are price setters' extent of indexation to past inflation<sup>3</sup> and the forward-looking component in the intertemporal IS curve, respectively;  $\kappa$  is the slope of the Philips curve;  $\sigma$  is the elasticity of intertemporal substitution in consumption;  $\rho$ ,  $\phi_{\pi}$ , and  $\phi_y$  are the smoothing coefficient and the long-run coefficients on inflation and the output gap in the monetary policy rule, respectively; and  $\epsilon_{\pi,t}$ ,  $\epsilon_{y,t}$ , and  $\epsilon_{R,t}$ are three structural disturbances following the AR(1) processes  $\epsilon_{x,t} = \rho_x \epsilon_{x,t-1} + \tilde{\epsilon}_{x,t}$ , for  $x=\pi$ , y, R, with  $\tilde{\epsilon}_{x,t} \sim N(0, \sigma_x^2)$ .

#### 2.1.1 Model solution under determinacy and indeterminacy

By defining the state vector as  $\xi_t \equiv [R_t, \pi_t, y_t, \pi_{t+1|t}, y_{t+1|t}, \epsilon_{R,t}, \epsilon_{\pi,t}, \epsilon_{y,t}]'$ , the vector collecting the structural shocks as  $\epsilon_t \equiv [\tilde{\epsilon}_{R,t}, \tilde{\epsilon}_{\pi,t}, \tilde{\epsilon}_{y,t}]'$ , and the vector of forecast errors as  $\eta_t \equiv [\eta_t^{\pi}, \eta_t^y]'$ —where  $\eta_t^{\pi} \equiv \pi_t - \pi_{t|t-1}$  and  $\eta_t^y \equiv y_t - y_{t|t-1}$ , the model can then be put into the 'Sims canonical form'<sup>4</sup>

$$\Gamma_0 \xi_t = \Gamma_1 \xi_{t-1} + \Psi \epsilon_t + \Pi \eta_t \tag{4}$$

where  $\Gamma_0$ ,  $\Gamma_1$ ,  $\Psi$  and  $\Pi$  are matrices conformable to  $\xi_t$ ,  $\epsilon_t$  and  $\eta_t$ .

In order to solve the model under both determinacy and indeterminacy, following Lubik and Schorfheide (2003) we exploit the QZ decomposition of the matrix pencil ( $\Gamma_0-\lambda\Gamma_1$ ). Specifically, given a pencil ( $\Gamma_0-\lambda\Gamma_1$ ), there exist matrices  $Q, Z, \Lambda$ , and  $\Omega$ such that  $QQ'=Q'Q=ZZ'=Z'Z=I_n$ ,  $\Lambda$  and  $\Omega$  are upper triangular,  $\Lambda=Q\Gamma_0Z$ , and  $\Omega=Q\Gamma_1Z$ . By defining  $w_t=Q'\xi_t$ , and by premultiplying (4) by Q, we have:

$$\begin{bmatrix} \Lambda_{11} & \Lambda_{12} \\ \hline 0 & \Lambda_{22} \end{bmatrix} \begin{bmatrix} w_{1,t} \\ w_{2,t} \end{bmatrix} = \begin{bmatrix} \Omega_{11} & \Omega_{12} \\ \hline 0 & \Omega_{22} \end{bmatrix} \begin{bmatrix} w_{1,t-1} \\ w_{2,t-1} \end{bmatrix} + \begin{bmatrix} Q_{1.} \\ \hline Q_{2.} \end{bmatrix} (\Psi \epsilon_t + \Pi \eta_t)$$
(5)

where the vector of generalised eigenvalues,  $\lambda$  (equal to the ratio between the diagonal elements of  $\Omega$  and  $\Lambda$ ) has been partitioned as  $\lambda = [\lambda'_1, \lambda'_2]'$ , with  $\lambda_2$  collecting all the explosive eigenvalues, and  $\Omega$ ,  $\Lambda$ , and Q have been partitioned accordingly. In

 $<sup>^{3}</sup>$ See e.g. Smets and Wouters (2003) and Christiano, Eichenbaum, and Evans (2005). The specific formulation we use herein is Smets and Wouters'.

 $<sup>^{4}</sup>$ See Sims (2002).

particular,  $Q_j$  collects the blocks of rows corresponding to the stable (j=1) and, respectively, unstable (j=2) eigenvalues. The explosive block of (5) can then be rewritten as

$$w_{2,t} = \Lambda_{22}^{-1} \Omega_{22} w_{2,t-1} + \Lambda_{22}^{-1} \left( \Psi_x^* \epsilon_t + \Pi_x^* \eta_t \right) \tag{6}$$

where  $\Psi_x^* = Q_2 \Psi$ , and  $\Pi_x^* = Q_2 \Pi$ . Given that  $\lambda_2$  is purely explosive, obtaining a stable solution to (4) requires  $w_{2,t}$  to be equal to 0 for any  $t \ge 0$ . This can be accomplished by setting  $w_{2,0}=0$ , and by selecting, for each t > 0, the forecast error vector  $\eta_t$  in such a way that  $\Psi_x^* \epsilon_t + \Pi_x^* \eta_t = 0$ .

Under determinacy, the dimension of  $\eta_t$  is exactly equal to the number of unstable eigenvalues, and  $\eta_t$  is therefore uniquely determined. Under indeterminacy, on the other hand, the number of unstable eigenvalues falls short of the number of forecast errors, and the forecast error vector  $\eta_t$  is therefore not uniquely determined, which is at the root of the possibility of sunspot fluctuations. Lubik and Schorfheide (2003), however, prove the following. By defining  $UDV'=\Pi_x^*$  as the singular value decomposition of  $\Pi_x^*$ , and by assuming that for each  $\epsilon_t$  there always exists an  $\eta_t$  such that  $\Psi_x^*\epsilon_t + \Pi_x^*\eta_t=0$  is satisfied, the general solution for  $\eta_t$  is given by

$$\eta_t = \left[ -V_{.1} D_{11}^{-1} U_{.1}' \Psi_x^* + V_{.2} M_1 \right] \epsilon_t + V_{.2} M_2 s_t^* \tag{7}$$

where  $D_{11}$  is the upper-left diagonal block of D, containing the square roots of the non-zero singular values of  $\Pi_x^*$  in decreasing order;  $s_t^*$  is a vector of sunspot shocks; and  $M_1$  and  $M_2$  are matrices whose entries are not determined by the solution procedure, and which basically 'index' (or parameterise) the model's solution under indeterminacy. Concerning  $M_1$  and  $M_2$  we follow Lubik and Schorfheide (2004), first, by setting  $M_2s_t^*=s_t$ , where  $s_t$  can therefore be interpreted as a vector of 'reduced-form' sunspot shocks. Second, we choose the matrix  $M_1$  in such a way as to preserve continuity of the impact matrices of the impulse-responses of the model at the boundary between the determinacy and the indeterminacy region. Specifically, let  $\theta$  be the parameters' vector, and let  $\Theta_I$  and  $\Theta_D$  be the sets of all the  $\theta$ 's corresponding to the indeterminacy and, respectively, to the determinacy regions. For every  $\theta \in \Theta_I$  we identify a corresponding vector  $\tilde{\theta} \in \Theta_D$  laying just on the boundary between the two regions.<sup>5</sup> By definition, the two impact matrices for the impulse-responses of the model conditional on  $\theta$  and  $\tilde{\theta}$  are given by

$$\frac{\partial \xi_t(\theta, M_1)}{\partial \epsilon_t} = \Psi^*(\theta) \cdot \Pi^*(\theta) V_{\cdot 1}(\theta) D_{11}^{-1}(\theta) U'_{\cdot 1}(\theta) \Psi^*_x(\theta) + \Pi^*(\theta) V_{\cdot 2}(\theta) M_1 \equiv (8)$$
$$\equiv B_1(\theta) + B_2(\theta) M_1 \tag{9}$$

<sup>5</sup>Specifically, for any  $[\phi_{\pi}, \phi_{y}]'$  such that  $\theta \in \Theta_{I}$ , we choose the vector  $[\tilde{\phi}_{\pi}, \tilde{\phi}_{y}]'$ , such that the resulting  $\tilde{\theta} \in \Theta_{D}$  lies just on the boundary between the two regions, by minimising the criterion  $\tilde{C} = [(\phi_{\pi} - \tilde{\phi}_{\pi})^{2} + (\phi_{y} - \tilde{\phi}_{y})^{2}]^{1/2}$ . It is important to stress that, in general, there is no clear-cut criterion for choosing a specific vector on the boundary. Minimisation of  $\tilde{C}$  is based on the intuitive notion of taking, as the 'benchmark'  $\tilde{\theta}$ , the one that is closest in vector 2-norm to  $\theta$ .

and, respectively,

$$\frac{\partial \xi_t(\tilde{\theta})}{\partial \epsilon_t} = \Psi^*(\tilde{\theta}) - \Pi^*(\tilde{\theta}) V_{\cdot 1}(\tilde{\theta}) D_{11}^{-1}(\tilde{\theta}) U'_{\cdot 1}(\tilde{\theta}) \Psi^*_x(\tilde{\theta}) \equiv B_1(\tilde{\theta})$$
(10)

where  $\Psi^*(\cdot) \equiv Q\Psi(\cdot)$ , and  $\Pi^*(\cdot) \equiv Q\Pi(\cdot)$ . We minimise the difference between the two impact matrices,  $B_1(\tilde{\theta})$ - $[B_1(\theta)+B_2(\theta)M_1]=[B_1(\tilde{\theta})-B_1(\theta)]-B_2(\theta)M_1$  by means of a least-squares regression of  $[B_1(\tilde{\theta})-B_1(\theta)]$  on  $B_2(\theta)$ , thus obtaining  $\tilde{M}_1=[B_2(\theta)'B_2(\theta)]^{-1}\times B_2(\theta)'[B_1(\tilde{\theta})-B_1(\theta)]$ .

The solution to (1)-(3) is now completely characterised. The forecast error  $\eta_t$  can be substituted into the law of motion for  $w_{1,t}$ ,

$$w_{1,t} = \Lambda_{11}^{-1} \Omega_{11} w_{1,t-1} + \Lambda_{11}^{-1} Q_{1.} \left( \Psi \epsilon_t + \Pi \eta_t \right)$$
(11)

thus obtaining, under both regimes, a VAR(1) representation for  $\xi_t$ ,

$$\xi_t = A_0 \xi_{t-1} + B_0 u_t \tag{12}$$

where  $u_t$  is vector standard white noise. Finally, the state-space representation of the model in terms of the three observable variables,  $R_t$ ,  $\pi_t$ ,  $y_t$ , implies the following observation equation

$$Y_t = C_0 \xi_t \tag{13}$$

with  $Y_t \equiv [R_t, \pi_t, y_t]'$  and  $C_0 = [I_3 \ 0_{3 \times (N_0 - 3)}]$ , where  $N_0$  is the dimension of the state vector. (Notice that, in terms of the canonical 'A-B-C-D' representation of a state-space form, the matrix  $D_0$  is here equal to  $D_0 = 0_{3 \times 3}$ .)

#### 2.1.2 The experiment's design

Our goal is to assess the performance of (structural) VARs conditional on a DGP in which neither luck (i.e., changes in the volatilities of the structural shocks), nor structural change (in the present case, changes in the non-policy parameters,  $\alpha$ ,  $\gamma$ ,  $\kappa$ ,  $\sigma$ , and all of the  $\rho_x$ 's), play *any* role whatsoever.

We therefore estimate (1)-(3)

- imposing indeterminacy for the pre-October 1979 period and determinacy for the period following the end of the Volcker stabilisation, by allowing for different values of ρ, φ<sub>π</sub>, and φ<sub>y</sub> across periods;
- imposing that  $\alpha$ ,  $\gamma$ ,  $\kappa$ ,  $\sigma$ , all of the  $\rho_x$ 's, and all of the  $\sigma_x^2$ 's, be identical across regimes. This is obtained by jointly estimating the two models for the pre-October 1979 and the post-Volcker stabilisation periods.

By showing that this DGP can replicate the key features of the Great Moderation, our results will illustrate, in the starkest possible way, that existing VAR evidence is *compatible* with the good policy explanation of the Great Moderation.

## 2.2 Bayesian estimation

We estimate (1)-(3) via Bayesian methods. The following two sub-sections describe our choices for the priors, and the Random-Walk Metropolis algorithm we use to get draws from the posterior.

#### 2.2.1 Priors

Following, e.g., Lubik and Schorfheide (2004) and An and Schorfheide (2006), all structural parameters are assumed, for the sake of simplicity, to be *a priori* independent from one another. The third column of Table 1 reports the parameters' prior densities, whereas the fourth and the fifth columns report two key objects characterising them, the mode and the standard deviation. Different from the vast majority of the papers in the literature, we calibrate the Gamma, inverse Gamma, and Beta prior densities in terms of the *mode* of the distribution, rather than in terms of the *mean* (specifically, we calibrate the densities so that our 'preferred values' for the parameters of interest are equal to the mode). The key reason for doing so is in order to give the maximal amount of prior weight to our 'preferred values', which, on the other hand, would not be the case if calibration were performed in such a way as to make the densities' means equal to such values.

#### 2.2.2 Getting draws from the posterior via Random-Walk Metropolis

We numerically maximise the log posterior—defined as  $\ln L(\theta|Y) + \ln P(\theta)$ , where  $\theta$  is the vector collecting the model's structural parameters,  $L(\theta|Y)$  is the likelihood of  $\theta$  conditional on the data, and  $P(\theta)$  is the prior—via simulated annealing (for a full description of the methodology, see Appendix A.1) We then generate draws from the posterior distribution of the model's structural parameters via the Random Walk Metropolis (henceforth, RWM) algorithm as described in, e.g., An and Schorfheide (2006). In implementing the RWM algorithm we exactly follow An and Schorfheide (2006, Section 4.1), with the single exception of the method we use to calibrate the covariance matrix's scale factor—the parameter c below—for which we follow the methodology described in Appendix D.3 of Benati (2008), which is briefly described in Appendix A.2 below.

Let then  $\hat{\theta}$  and  $\hat{\Sigma}$  be the mode of the maximised log posterior and its estimated Hessian, respectively.<sup>6</sup> We start the Markov chain of the RWM algorithm by drawing  $\theta^{(0)}$  from  $N(\hat{\theta}, c^2 \hat{\Sigma})$ . For s = 1, 2, ..., N we then draw  $\tilde{\theta}$  from the proposal distribution  $N(\theta^{(s-1)}, c^2 \hat{\Sigma})$ , accepting the jump (i.e.,  $\theta^{(s)} = \tilde{\theta}$ ) with probability min  $\{1, r(\theta^{(s-1)}, \theta^{(s-1)}, \theta^{(s-1)})\}$ , and rejecting it (i.e.,  $\theta^{(s)} = \theta^{(s-1)})$  otherwise, where

$$r(\theta^{(s-1)}, \theta | Y) = \frac{L(\theta | Y) P(\theta)}{L(\theta^{(s-1)} | Y) P(\theta^{(s-1)})}$$

<sup>&</sup>lt;sup>6</sup>We compute  $\hat{\Sigma}$  numerically as in An and Schorfheide (2006).

We run a burn-in sample of 200,000 draws which we then discard. After that, we run a sample of 200,000 draws, keeping every draw out of 100 in order to decrease the draws' autocorrelation, thus ending up with a sample of 2,000 draws.

Table 1 reports the modes and the 90%-coverage percentiles of the distributions of the model's structural parameters.

# 3 Theoretical Properties of the Estimated Data-Generation Process

In this Section we explore the theoretical properties of the estimated data-generation process under the two regimes, by analysing the structural VAR(MA) representations of the model, the structural innovations' theoretical impact matrices and impulse-response functions, the VAR(MA)'s reduced-form innovation variances, and the series' theoretical variances under determinacy and indeterminacy. By focusing on the theoretical properties of the DGP, we will therefore show that the ability of Clarida *et al.*'s 'indeterminacy hypothesis' to replicate the broad features of the Great Moderation as a consequence of a shift in monetary policy has nothing to do with estimation issues—sample length, choice of the lag order, etc.—but it rather holds *in population*.

In order to make the exposition clearer, however, it is useful to start with the model's theoretical equivalent minimal state-space representation under the two regimes.

## 3.1 The equivalent minimal state-space representations of the model under the two regimes

Conditional on the estimates reported in Table 1, the theoretical state-space representations of the model under the two regimes can easily be computed. By applying MATLAB's routine **ss.m** to the two state-space forms we then obtain the two equivalent minimal state-space representations (henceforth, EMSSR) of the model,

$$\xi_t^M = \begin{bmatrix} 0.68 & -0.31 & 0.28 & -0.09 & 0.09 & 0.01 & 0.00 \\ 0.30 & 0.61 & 0.07 & 0.11 & -0.10 & -0.01 & 0.00 \\ 0.42 & 0.04 & 0.58 & -0.13 & 0.04 & 0.00 & 0.00 \\ -0.32 & 0.08 & -0.12 & 0.29 & -0.24 & 0.01 & 0.00 \\ 0.06 & -0.06 & -0.05 & -0.31 & 0.53 & 0.05 & 0.00 \\ 0.19 & -0.12 & -0.25 & -0.20 & 1.10 & 0.43 & 0.02 \\ 0.71 & -0.47 & 0.66 & 0.34 & 0.00 & -0.03 & 0.40 \end{bmatrix} \xi_{t-1}^M + \xi_{t-1}$$

$$H \begin{bmatrix} 0.55 & 0.15 & 0.3968 & -0.11 \\ 0.15 & 0.57 & 0.5881 & 0.16 \\ 0.47 & -0.08 & 0.24 & -0.08 \\ 0.00 & -0.08 & 0.24 & -0.08 \\ -0.27 & 0.30 & -0.03 & -0.11 \\ -0.15 & 1.28 & -0.29 & -0.03 \\ 1.54 & 0.09 & -0.30 & 0.03 \end{bmatrix} u_t$$
(14)  
$$H = \begin{bmatrix} 0.08 & -0.67 & -0.67 & 0.22 & -0.21 & -0.02 & 0.00 \\ 0.00 & -0.06 & 0.09 & 0.38 & 0.35 & -0.64 & -0.56 \\ -0.01 & -0.16 & 0.27 & 0.22 & -0.19 & 0.64 & -0.64 \end{bmatrix} \xi_t^M$$
(15)

under indeterminacy, and

Y

$$\xi_t^M = \begin{bmatrix} 0.78 & 0.16 & 0.18 & -0.20 & -0.02 & 0.00 \\ 0.36 & 0.51 & 0.02 & -0.15 & -0.02 & 0.00 \\ -0.19 & 0.00 & 0.16 & -0.07 & 0.02 & 0.00 \\ 0.14 & -0.11 & -0.26 & 0.46 & 0.04 & 0.00 \\ 0.24 & -0.40 & 0.09 & 0.73 & 0.39 & 0.02 \\ -0.11 & 0.42 & 0.72 & -0.44 & -0.08 & 0.40 \end{bmatrix} \\ \xi_{t-1}^M + \begin{bmatrix} -0.70 & 0.51 & -0.36 \\ -0.14 & -0.26 & -0.05 \\ 0.08 & -0.20 & -0.16 \\ 0.22 & 0.42 & 0.06 \\ 0.27 & 1.03 & -0.81 \\ 0.04 & -0.92 & -1.03 \end{bmatrix} u_t$$

$$Y_t = \begin{bmatrix} -0.67 & -0.66 & 0.23 & -0.21 & -0.02 & 0.00 \\ -0.04 & 0.08 & 0.38 & 0.33 & -0.64 & -0.57 \\ -0.17 & 0.26 & 0.20 & -0.18 & 0.64 & -0.65 \end{bmatrix} \\ \xi_t^M \tag{17}$$

under determinacy, where  $\xi_t^M$  is the state vector in the EMSSR,<sup>7</sup> and  $Y_t$  is still equal to  $Y_t \equiv [R_t, \pi_t, y_t]'$ . A comparison between (14)-(15) and (16)-(17) immediately highlights a fundamental difference between the two regimes. Whereas under determinacy the EMSSR has six states, under indeterminacy it has *seven*. (The fact that, under indeterminacy, the EMSSR possesses an additional state variable compared with the determinacy regime was first shown in a previous version of this paper that was presented at the New York FED on September 27th, 2006. The slides of the presentation are available from either of us, while the paper is available either from us, or from the New York FED.) An important point to stress is that the presence of an additional state variable under indeterminacy has *nothing* to do with the presence of a sunspot shock. Indeed, first, it can be easily shown that this feature of the DGP remains unchanged even if we set the variance of the sunpot shock to zero. Second, and more fundamentally, given that the sunspot shock is pure white noise, on strictly logical/mathematical grounds it *cannot* belong to the state vector, so that the additional state under indeterminacy *ought* to be something else.<sup>8</sup> Rather, the presence

<sup>&</sup>lt;sup>7</sup>In general, the states of the EMSSR generated by MATLAB's routine ss.m do not have any intrinsic meaning whatsoever, as they are simply linear combinations of the states of the original state-space form.

<sup>&</sup>lt;sup>8</sup>For an extensive discussion of what the additional state variable under indeterminacy exactly is, see the Appendix B.

of an additional state under indeterminacy is a *general property* of New Keynesian models under this regime.

#### 3.2 The VARMA representations

Such finding has the following important implication. Since, once the EMSSR has been appropriately rotated,<sup>9</sup> three of the states in  $\xi_t^M$  are equal to the three structural disturbances, this automatically implies that under determinacy, with six states in the EMSSR, the model possesses a VAR representation in  $R_t$ ,  $\pi_t$ , and  $y_t$ . Indeed, applying to the EMSSR of the model under determinacy a MATLAB code for computing the finite-order VAR representation of a state-space form,<sup>10</sup> we obtain the VAR(2) representation

$$Y_{t} = \underbrace{\begin{bmatrix} 1.21 & 0.01 & 0.14 \\ -0.03 & 0.47 & 0.07 \\ -0.11 & -0.05 & 1.02 \end{bmatrix}}_{B_{1}^{DET}} Y_{t-1} + \underbrace{\begin{bmatrix} -0.32 & 0.01 & -0.05 \\ 0.02 & -0.02 & -0.02 \\ 0.08 & 0.00 & -0.23 \end{bmatrix}}_{B_{2}^{DET}} Y_{t-2} + v_{t},$$
with  $\operatorname{Var}(v_{t}) = \begin{bmatrix} 0.437 & 0.259 & -0.168 \\ 0.259 & 1.184 & 0.039 \\ -0.168 & 0.039 & 1.025 \end{bmatrix}$  (18)

Under indeterminacy, on the other hand, the very same logic implies that, with one additional state variable in the EMSSR, the model does not possess a pure VAR representation in  $R_t$ ,  $\pi_t$ , and  $y_t$ , but rather a VARMA one, with a small moving-average component. Figure 1 illustrates this, by showing the evolution of the coefficients of the theoretical VAR( $\infty$ ) implied by the VARMA representation of the model under indeterminacy, as a function of the lag. A simple illustration of the speed of decay towards zero of the coefficients of the VAR( $\infty$ ) representation of the model is provided by the evolution of the maximum among the absolute values of the elements of the AR matrices of the VAR( $\infty$ ). At the first three lags, such maximum is equal to 1.0561, 0.2463, and 0.0513, respectively, whereas at lags 10 and 20 it decreases to 0.014 and 2.2E-3, respectively, and at lags 50, 75, and 100 it further declines to 8.6E-6, 8.4E-8, and 1.6E-9, respectively. Finally, the covariance matrix of reduced-form innovations of the VARMA representation of the model obtained when setting the sunspot shock to zero is given by

$$\operatorname{Var}(v_t) = \begin{bmatrix} 0.472 & 0.468 & 0.022\\ 0.468 & 1.407 & 0.147\\ 0.022 & 0.147 & 1.306 \end{bmatrix}$$
(19)

<sup>9</sup>In general, state-space forms are unique up to a rotation—see the discussion in Appendix B. <sup>10</sup>The code has been kindly supplied by Juan Rubio-Ramirez.

# 3.3 Impulse-response functions of the model under the two regimes

Figure 2 plots the theoretical impulse-response functions (henceforth, IRFs) of the model under the two regimes. As the figure shows, in most cases the differences between the the IRFs under determinacy and indeterminacy is far from dramatic—in a few cases it is essentially trivial—thus pointing towards intrinsic difficulties in distinguishing between the two regimes based on an analysis of the IRFs.

# 4 Replicating the Great Moderation

## 4.1 Volatility decreases in population

A comparison between the diagonal elements of (19) and (18) shows that the shift in the systematic component of monetary policy associated with the move from indeterminacy to determinacy is *sufficient* to generate decreases in the theoretical innovation variances for all series as a simple implication of the Lucas (1976) critique, and without any need of sunspot shocks. Whereas earlier contributions have interpreted decreases in reduced-form innovation variances as *prima facie* evidence in favor of good luck, and against good policy, our results clearly show such interpretation to be unwarrented. Further, such policy shift is associated with decreases in the standard deviations of both inflation and the output gap, from 1.32 to 1.23 and from 2.11 to 1.88 respectively, whereas the standard deviation of the interest rate slightly increases from 1.65 to 1.72. With the estimated standard deviation of sunspot shocks, the model exhibits clear decreases in both variances and innovation variances when moving from indeterminacy to determinacy. In particular, the theoretical standard deviations of  $R_t$ ,  $\pi_t$ , and  $y_t$  under indeterminacy become equal to 4.41, 4.38, 2.48, respectively, thus highlighting the ability of the estimated DGP to broadly replicate the generalised decline in overall macroeconomic volatility associated with the Great Moderation.

## 4.2 Generating 'Great Inflations' and 'Great Moderations'

Figure 3 provides a stark illustration of the ability of Clarida *et al.*'s 'indeterminacy hypothesis' to replicate the transition from the Great Inflation to the Great Moderation *uniquely* as a result of improved monetary policy, by plotting a single stochastic simulation of length T=100 for both regimes from the estimated model. As the figure shows, the first part of the sample exhibits what, at first blush, clearly looks like a 'Great Inflation' episode, with both inflation and the nominal rate displaying wide and persistent swings, reaching peaks around twelve-thirteen per cent.<sup>11</sup> In the sec-

<sup>&</sup>lt;sup>11</sup>This is conceptually in line with Clarida *et al.*'s (2000) Figure V on page 172.

ond half of the sample, on the other hand, fluctuations are much more subdued, and do not exhibit any persistent deviation from equilibrium.

## 5 Can Structural VAR Methods Uncover the Truth?

When applied to the estimated model, can structural VAR methods uncover the authentic causes of the changes in the DGP across regimes? As this section shows, the answer is unfortunately 'No'.

#### 5.1 Theoretical structural policy counterfactuals

We start by analysing *theoretical* policy counterfactuals, i.e. counterfactuals based on the theoretical VAR representations of the model under the two regimes. By showing that structural VAR methods fail to correctly capture the truth *in population*, we will therefore illustrate in the starkest possible way that the problems discussed in the present work have nothing to do with *estimation* issues—i.e, lag order selection and the like—but rather point towards fundamental weaknesses of the structural VAR approach for the present purposes.

As we pointed out in Section 3.2, the model possesses a VAR(2) representation under determinacy, and a VARMA one with a small moving-average component under indeterminacy. We start by approximating the VAR( $\infty$ ) implied by the VARMA representation under indeterminacy with a VAR(100),<sup>12</sup> and we augment the VAR(2) under determinacy with 98 further AR matrices equal to  $0_{3\times3}$ . Based on the structural shocks' theoretical impact matrices for the two regimes, we then put the two theoretical VARs into the corresponding structural VAR forms,

$$A_{0,IND}^{-1}Y_t = \tilde{B}_1^{IND}Y_{t-1}... + \tilde{B}_{100}^{IND}Y_{t-100} + \epsilon_t$$
(20)

$$A_{0,DET}^{-1}Y_t = \tilde{B}_1^{DET}Y_{t-1}... + \tilde{B}_{100}^{DET}Y_{t-100} + \epsilon_t$$
(21)

where  $\tilde{B}_j^x = A_{0,x}^{-1} B_j^x$ , with x = IND, DET (with IND for 'indeterminacy' and DET'determinacy'),  $A_{0,x}^{-1}$  being the impact matrix of the three structural shocks ( $\tilde{\epsilon}_{R,t}, \tilde{\epsilon}_{\pi,t}, \tilde{\epsilon}_{y,t}$ ) at zero, and j = 0, 1, ..., 100. A crucial point to stress is that, since we are here working with the impact matrices of the three structural shocks, we are implicitly setting the variance of the sunspot shock to zero—to put it differently, the version of the model we are working in this sub-section is the one *without* sunspots. By illustrating the failure of structural VAR-based policy counterfactuals to capture the truth even in the absence of sunspot shocks, we will therefore show that the problems discussed in this sub-sections have nothing to do with the presence of sunspot shocks, and rather pertain to a fundamental problem of VAR-based policy counterfactuals.

<sup>&</sup>lt;sup>12</sup>As we pointed out in Section 3.2, at lag 100 the maximum among the absolute values of the elements of the AR matrix of the VAR( $\infty$ ) is of an order of magnitude of 10<sup>-11</sup>, which implies that all lags beyond the 100<sup>th</sup> can safely be ignored.

After switching the structural monetary rules in the two structural VARs—i.e., the first equations in (20) and (21)—we convert the counterfactual structural VARs we thus obtain into corresponding counterfactual reduced-form VARs, from which theoretical counterfactual standard deviations for the three series can trivially be computed. As we pointed out in Section 4.1, under determinacy the true theoretical standard deviations are equal to 1.72, 1.23, and 1.88 respectively. Quite strikingly, the theoretical counterfactual standard deviations we obtain by imposing the structural monetary rule corresponding to the indeterminacy regime onto the structural VAR for the determinacy regime<sup>13</sup> are equal to 1.21, 1.22, and 1.92, thus implying a volatility *decrease* for two series out of three, rather than an increase. Results from the alternative counterfactual in which we impose the structural monetary rule corresponding to the determinacy regime onto the structural VAR for the indeterminacy regime<sup>14</sup> are equally striking. Whereas, under indeterminacy, the true standard deviations of the three series are equal to 4.41, 4.38, and 2.48, respectively, the counterfactual standard deviations we obtain via the policy switch are equal to 16.84, 9.47, and 2.12 respectively, thus implying a volatility *increase* for two series out of three, rather than a decrease, thus highlighting, once again, the failure of structural VAR-based policy counterfactuals to capture the truth.

#### 5.1.1 Why do structural-VAR based policy counterfactuals fail?

What is going on here? Why do structural-VAR based policy counterfactuals fail so badly in population? The reason is not difficult to grasp, and is the following. When performing counterfactual simulations in structural VARs, the *implicit presumption* is that switching the VAR's estimated structural policy rules should provide a reasonable approximation to the *authentic* policy switch, i.e. the one between the Taylor rules in the underlying DSGE model. As our results show, however, such presumption is, in general, unwarranted, the key reason being that what is structural as defined by the underlying DSGE model bears no clear-cut connection with what is *defined* as 'structural' by the structural VAR form of the very same DSGE model, i.e. equations (20) and (21). The difference between these two notions of what is structural is at the root of the problem here, and cannot therefore be 'fixed', being rather a fundamental shortcoming of policy counterfactuals based on structural VARs.

This problem of structural-VAR based policy counterfactuals is extensively analysed by Benati (2007). Taking a standard New Keynesian model as DGP, he explores the conditions under which SVAR-based policy counterfactuals may provide a reasonably good approximation to the 'authentic' policy switch, i.e. the one between the Taylor rules in the New Keynesian model. As he shows, such conditions are extremely restrictive, to the point that structural VAR-based policy counterfactuals should in general be regarded as unreliable. The logical corollary of this result is that all results

<sup>13</sup>I.e, the counterfactual in which we 'bring Arthur Burns into the post-Volcker stabilisation era'.

<sup>&</sup>lt;sup>14</sup>I.e, we 'bring Paul Volcker or Alan Greenspan back in time'.

from structural VAR-based policy counterfactuals should be regarded as *uninformative* for the issue of properly assessing the role played by monetary policy in fostering the Great Moderation.

#### 5.2 Impulse-response functions

Let's now turn to IRFs. Little change over time in estimated IRFs to an identified monetary policy shock has been traditionally regarded as evidence in favor of good luck, and against good policy. As we will now show, such evidence is, once again, uninformative for the issue of deciding the role played by monetary policy in fostering the Great Moderation.

Figure 4 shows, for the two regimes, the medians and the 90%-coverage percentiles of the distributions of the estimated IRFs to a 100 basis points monetary policy shock, based on 1,000 stochastic simulations. Specifically, for each of the 1,000 simulations

(i) we generated artificial data under the two regimes from the estimated DGP and we estimated reduced-form VARs exactly as in Section 4.2, choosing the lag order based on the AIC.

(*ii*) Based on the VARs' estimated covariance matrices, we estimate the structural impact matrices for the two regimes by imposing the true theoretical sign restrictions<sup>15</sup> via the procedure introduced by Rubio-Ramirez, Waggoner, and Zha (2005). We integrate out rotation uncertainty by computing, for each of the 1,000 stochastic simulations, 1,000 impact matrices satisfying the sign restrictions, and then taking the average among them.

(ii) Based on the estimated VARs and the structural impact matrices, we compute the IRFs to a 100 basis points shock to the interest rate.

Two main findings emerge from the Figure. First, the distributions of the estimated IRFs under indeterminacy are much wider than those under determinacy. This is especially clear for inflation and the nominal rate, much less so for the output gap, and finds its origin in the greater persistence exhibited by the system under indeterminacy. Second, for none of the series it is possible to reject the null that the IRFs have remained unchanged across regimes.

Although little change in estimated IRFs to a monetary policy shock have routinely been interpreted as evidence in favor of good luck and against good policy, these results show that this interpretation is unwarranted.

<sup>&</sup>lt;sup>15</sup>The reason for using sign restrictions is that *any* other identification scheme (e.g., Cholesky) would be false in population—in the sense of being inconsistent with the true data generation process—so that its results could not be meaningfully interpreted in any way. This is conceptually in line with Canova and Pina (2005).

# 6 Some Criticisms of Our Analysis, and Our Rebuttals

### 6.1 Canova's criticism

Fabio Canova has circulated a note which is critical of the present work.<sup>16</sup> He summarises his main objections as follows:<sup>17</sup>

'[N]one of the problems highlighted by Benati and Surico have to do with VAR methods, *per se.* It is the choice of experimental design, failure to recognize the presence of omitted variables in one regime and the choice of relatively small sample size which drive the results they obtain.'

In what follows we show that most of Canova's arguments are wrong, and none of them affects our conclusions.

#### 6.1.1 The additional state variable under indeterminacy

As we pointed out, the minimal state-space representation of the model has six state variables under determinacy, and *seven* under indeterminacy. Canova (2006) states that the additional state variable under indeterminacy is  $\pi_{t+1|t}$  (i.e. expected inflation at time t+1, conditional on information available at time t),<sup>18</sup> and based on this conjecture makes two points.

- If you augment the VAR in inflation, the output gap and the nominal interest rate with expected inflation, the augmented VAR may uncover the true causes of change in the DGP.
- If you integrate out the presence of the unobserved state variable under indeterminacy, the variances of the series are much lower, so that the model cannot replicate the Great Moderation (see footnote 1, and the variances reported in Table 1 of Canova's comment).

Concerning the first point, Canova and Gambetti (2007) take Canova's conjecture as the rationale for including measures of expected inflation in their (fixed-coefficient and time-varying parameters) VARs. Given that, overall, their results are largely unaffected by the inclusion of expected inflation measures, they conclude that their

 $<sup>^{16}\</sup>mathrm{See}$  Canova (2006).

 $<sup>^{17}\</sup>mathrm{See}$  Canova (2006, page 12, second paragraph).

<sup>&</sup>lt;sup>18</sup>To be precise, Canova (2006) is not explicit about what he actually means by 'expected inflation', which he just labels as  $\pi_t^e$ . Although, in principle, he could mean either  $\pi_{t+1|t}$  or  $\pi_{t|t-1}$ , the latter can be ruled out on strictly logical grounds. Given that  $\pi_{t|t-1}$  does not belong to the state vector of the *original* state-space form,  $\xi_t$ , there is simply *no way* that it can belong to the state vector of the EMSR,  $s_t$ . The only possibility left is therefore  $\pi_{t+1|t}$ .

evidence runs against Clarida et al.'s (2000) indeterminacy hypothesis about the sources of the Great Moderation.

The problem with Canova's and Canova and Gambetti's position is that, as we demonstrate mathematically in Appendix B, the VAR representation they consider is only one among an infinity of admissible VAR representations for the economy, so that their evidence is ultimately uninformative for the issue at hand. Another admissible representation, for example, has expected inflation replaced by the expected output gap, or by any linear (convex or non-convex) combination of the two. As a simple corollary, this automatically implies that a negative result—like the one obtained by Canova and Gambetti (2007)—is not telling us, in principle, anything: to be informative, the negative outcome should be obtained for all the infinite, possible representations of the economy. What truly is informative under these circumstances, on the other hand, is a positive result, i.e. the finding that a particular variable which is admissible as the fourth element in the VAR representation of the model beyond  $R_t$ ,  $\pi_t$ , and  $y_t$  does indeed Granger-cause the first three variables, and/or produces significant changes in the results.

As for the second issue raised by Canova (2006), the crucial point to stress is that the presence of an additional state variable under indeterminacy is an integral part of the *intrinsic dynamics* of the system under that regime, so that it is simply not possible to meaningfully talk of the dynamics of inflation, the interest rate, and the output gap under indeterminacy 'controlling' for the effect of this additional state. To make this point even clearer, consider the following simple example. Under *determinacy* the dynamics of the economy only depends on three states, inflation, the output gap, and the interest rate. Following Canova's logic, we could say that 'if we control for (i.e, integrate out) the influence of the interest rate, the variance of inflation under determinacy would be lower'. The key problem with this is that 'controlling for the influence of the interest rate—exactly like the additional unobserved state variable under indeterminacy—*is not exogenous*, but it is rather *endogenous* to the system. You might legitimately want to control for fluctuations in exogenous driving processes, but controlling for the influence of endogenous variables is meaningless.

#### 6.1.2 The experimental design

Our paper considers a world in which the Great Moderation is exclusively driven by improved monetary policy, and then asks: 'When applied to the simulated data, are the VAR methods used in earlier contributions capable of delivering the true answer of good policy?'. Canova (2006), on the other hand, is concerned with a completely different question: 'Is it possible to specify a DSGE model and a policy shift for which VAR analysis would uncover the true change in the DGP?<sup>19</sup> Unfortunately, this is

<sup>&</sup>lt;sup>19</sup> 'If one is interested in measuring the ability of VAR methods to answer the questions of interest, one should also design an experiment where only the lagged coefficients change. [...] To have

irrelevant for the issue at hand: the fact that I can conceive circumstances under which a specific econometric methodology performs well does not tell me anything about its performance conditional on the only DGP that truly matters, i.e. the one which is out there. The reason is very simple: reality is what it is, and you can't 'choose' it to suit the econometric methodologies that you like. As a consequence, in order to be reasonably confident that the results produced by a specific econometric methodology are sufficiently reliable, such methodology must be shown to perform well conditional on a wide range of plausible data-generation processes. As this paper has shown, however, if the truth for the post-WWII United States is reasonably well described, to a first approximation, by Clarida et al.'s (2000) 'indeterminacy hypothesis', VAR methods as they have been implemented so far may well point towards the incorrect conclusion of good luck. And this automatically implies that existing VAR evidence is uninformative for the issue at hand: if I always get good luck irrespective of what the truth is, such result is quite obviously uninformative for the issue at hand. The notion that a researcher may therefore pick and choose a specific DGP in order to show that, conditional on that DGP, his/her favorite econometric methodology performs well does therefore not appear to us a meaningful way of discriminating between what works and what doesn't work.

#### 6.1.3 VARs do have problems

As we already pointed out, one of Canova's key contentions is that '[...] none of the problems highlighted by Benati and Surico have to do with VAR methods, per se.' On this, we very much disagree: VAR methods do have problems which are intrinsic to this methodology. As we stressed in the introduction, indeed, there are two key dimensions along which VAR analysis turns out to be misleading within the present context. First, in general, changes in the coefficients of the monetary policy rule of the DSGE model exert their impact on *both* the coefficients of the VAR representation of the model, and the elements of the VAR's covariance matrix of reduced-form innovations. Here there's not much that one can quibble with this: the fact that a change in the parameters of the Taylor rule also exerts an impact on the the VAR's covariance matrix is nothing but a trivial consequence of the Lucas (1976) critique, and as such it is not possible to argue with it. Second, in general, changes in the monetary policy rule of a structural VAR bear no clear-cut relationship with changes in the parameters of the monetary policy rule in the underlying DSGE model. To put it differently, there appears to be a fundamental disconnect between what is structural within a DSGE model, and what is *defined* as structural based on the structural VAR representation implied by the very same DSGE model. This is a fundamental *conceptual* weakness of the structural VAR methodology, so that

a design with the required features, one should study within regime changes when the nominal interest rate reacts to *lagged* output gap and *lagged* inflation since, by construction, the matrix of impact coefficients is fixed across regimes while the matrix of lagged coefficients changes' (Canova, 2006, page 7, third paragraph).

Canova's contention that such problems have nothing to do with VAR methods *per* se is simply incorrect. This is a problem that lies at the very heart of the structural VAR methodology. It is also to be noticed that—as extensively discussed by Benati (2007)—this has nothing to do with Lucas critique-type problems, for the simple reason that, whereas the Lucas critique's key theme is the difference between what is structural and what is reduced-form, the fundamental issue here is the difference between what *truly is* structural, as defined by the DSGE model, and what is *labelled* as structural within the structural VAR representation implied by the very same DSGE model.

#### 6.1.4 The sample size

The original version of the paper Canova refers to was entirely based on stochastic simulations. Although the present version's analysis is done almost entirely in population—i.e., it is based on theoretical quantities—it is instructive to tackle this issue too.

In the original version we simulated the New-Keynesian model using 100 observations per regime, which, at the quarterly frequency, corresponds to 25 years of data. VAR analyses of the Great Moderation begin the sample period either at the end of the 1950s or at the beginning of the 1960s and typically split the full sample around October 1979. The sample selection implies that the inference drawn on the VAR estimates is based on 20 years of data for the 'bad policy' regime, and 20-25 years for the 'good policy' regime. Our choice of 100 observations was therefore *perfectly* in line with the number of data points available to the econometrician—indeed, this was *precisely* the reason why we chose it!

But at the end of the day this issue is irrelevant, as the present analysis is almost entirely based

### 6.2 A further criticism

A further criticism has been offered, so far, of the present analysis. The work of Clarida, Gali, and Gertler (2000) and Lubik and Schorfheide (2004), upon which the present analysis is based, treats the shift from the first to the second regime as 'once-and-for-all', rather than as an ongoing process. As pointed out, e.g., by Davig and Leeper,<sup>20</sup> '[o]nce-and-for-all shifts, by definition, are unanticipated, yet once the shift occurs, agents are assumed to believe the new regime is permanent and alternative regimes are impossible. But if regime has changed, then regime can change; knowing this, private agents will ascribe a probability distribution to regimes. Expectations formation and, therefore, the resulting equilibria will reflect agents' beliefs that regime change is possible.' The key point to understand, here, is that the problems highlighted in this paper have nothing to do with the specific way

 $<sup>^{20}</sup>$ See Davig and Leeper (2007).

in which policy changes are modelled—'once-and-for-all', as opposed to an ongoing process—and as a result should be expected to carry over, as a simple matter of logic, even within more realistic settings like those investigated by Davig and Leeper (2007).

# 7 Conclusions

Most analyses of the U.S. Great Moderation have been based on structural VAR methods, and have consistently pointed towards good luck as the main explanation for the greater macroeconomic stability of recent years. Based on an estimated New-Keynesian model in which the only sources of change are the move from passive to active monetary policy, and the presence of sunspots under indeterminacy, we show that VARs may misinterpret good policy for good luck. In particular, the estimated DGP exhibits decreases in population in both variances and innovation variances for all series. Policy counterfactuals based on the theoretical structural VAR representations of the model under the two regimes fail to capture the truth, whereas impulse-response functions to a monetary policy shock exhibit little change across regimes. Since these results are in line with those found in the structural VAR-based literature on the Great Moderation, our analysis suggests that existing VAR evidence is compatible with the 'good policy' explanation of the Great Moderation.

## References

- AN, S., AND F. SCHORFHEIDE (2006): "Bayesian Analysis of DSGE Models," *Econometric Reviews*, forthcoming.
- BENATI, L. (2007): "How Reliable Are Structural VAR-Based Policy Counterfactuals?," European Central Bank, in progress.
- BENATI, L. (2008): "Investigating Inflation Persistence Across Monetary Regimes," *Quarterly Journal of Economics*, forthcoming.
- BOIVIN, J., AND M. GIANNONI (2006): "Has Monetary Policy Become More Effective?," The Review of Economics and Statistics, 88, 445–462.
- "You F. (2006): Can Use VARs for Structural CANOVA, Analy-VARs Moderation," ses. А Comment and the Great at:  $\mathrm{to}$ http://www.crei.cat/people/canova/papers.html.
- CANOVA, F., AND L. GAMBETTI (2007): "Do Inflation Expectations Matter? The Great Moderation Revisited," Universitat Pompeu Fabra, mimeo.
- CANOVA, F., AND J. PINA (2005): "What VAR Tell Us About DSGE Models," in Diebolt, C. and Kyrtsou, C. New Trends In Macroeconomic, Springer Verlag.
- CHRISTIANO, L., M. EICHENBAUM, AND C. EVANS (2005): "Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy," *Journal of Political Economy*, 113(1), 1–45.
- CLARIDA, R., J. GALI, AND M. GERTLER (2000): "Monetary Policy Rules and Macroeconomic Stability: Evidence and Some Theory," *Quarterly Journal of Economics*, CXV(1), 147–180.
- CORANA, A., M. MARCHESI, C. MARTINI, AND S. RIDELLA (1987): "Minimizing Multimodal Functions of Continuous Variables with the Simulated Annealing Algorithm," *ACM Transactions on Mathematical Software*, 13, 262–80.
- DAVIG, T., AND E. M. LEEPER (2007): "Generalising the Taylor Principle," *American Economic Review*, forthcoming.
- GAMBETTI, L., E. PAPPA, AND F. CANOVA (2006): "The Structural Dynamics of US Output and Inflation: What Explains the Changes?," *Journal of Money*, *Credit, and Banking*, forthcoming.
- GOFFE, W. L., G. FERRIER, AND J. ROGERS (1994): "Global Optimization of Statistical Functions with Simulated Annealing," *Journal of Econometrics*, 60,(1/2), 65–99.



- KIM, C. J., AND C. NELSON (1999): "Has the U.S. Economy Become More Stable? A Bayesian Approach Based on a Markov-Switching Model of the Business-Cycle," *Review of Economics and Statistics*, 81, 608–616.
- LEEPER, E. (1991): "Equilibria Under 'Active' and 'Passive' Monetary and Fiscal Policies," *Journal of Monetary Economics*, 27, 129–147.
- LUBIK, T., AND F. SCHORFHEIDE (2003): "Computing Sunspot Equilibria in Linear Rational Expectations Models," *Journal of Economic Dynamics and Control*, 28(2), 273–285.
- (2004): "Testing for Indeterminacy: An Application to U.S. Monetary Policy," *American Economic Review*, 94(1), 190–217.
- LUCAS, R. E. (1976): "Econometric Policy Evaluation: A Critique," Carnegie-Rochester Conference Series on Public Policy, 1, 19–46.
- MCCONNELL, M., AND G. PEREZ-QUIROS (2000): "Output Fluctuations in the United States: What Has Changed Since the Early 1980s?," *American Economic Review*, 90, 1464–1476.
- MOJON, B. (2007): "Monetary Policy, Output Composition, and the Great Moderation," *Federal Reserve Bank of Chicago Working Paper 2007-07*.
- PRIMICERI, G. E. (2005): "Time Varying Structural Vector Autoregressions and Monetary Policy," *The Review of Economic Studies*, 72, 821–852.
- RUBIO-RAMIREZ, J., D. WAGGONER, AND T. ZHA (2005): "Regime Changes in the Euro Area," Federal Reserve Bank of Atlanta, mimeo.
- SIMS, C. (2002): "Solving Linear Rational Expectations Models," Computational Economics, 20, 1–20.
- SIMS, C., AND T. ZHA (2006): "Were There Regime Switches in U.S. Monetary Policy?," American Economic Review, 96(1), 54–81.
- SMETS, F., AND R. WOUTERS (2003): "An Estimated Stochastic Dynamic General Equilibrium Model of the Euro Area," Journal of the European Economic Association, 1(5), 1123–1175.
- STOCK, J., AND M. WATSON (2002): "Has the Business Cycle Changed and Why?," in M. Gertler and K. Rogoff, eds. (2003) NBER Macroeconomics Annuals 2002.



# A Two Technical Aspects of the Bayesian Estimation Procedure

This appendix discusses in detail two technical aspects the Bayesian estimation procedure.

### A.1 Numerical maximisation of the log posterior

We numerically maximise the log posterior—defined as  $\ln L(\theta|Y) + \ln P(\theta)$ , where  $\theta$  is the vector collecting the model's structural parameters,  $L(\theta|Y)$  is the likelihood of  $\theta$  conditional on the data, and  $P(\theta)$  is the prior—via simulated annealing. Following Goffe, Ferrier, and Rogers (1994) we implement simulated annealing via the algorithm proposed by Corana, Marchesi, Martini, and Ridella (1987), setting the key parameters to  $T_0=100,000, r_T=0.9, N_t=5, N_s=20, \epsilon=10^{-6}, N_{\epsilon}=4$ , where  $T_0$  is the initial temperature,  $r_T$  is the temperature reduction factor,  $N_t$  is the number of times the algorithm goes through the  $N_s$  loops before the temperature starts being reduced,  $N_s$  is the number of times the algorithm goes through the function before adjusting the stepsize,  $\epsilon$  is the convergence (tolerance) criterion, and  $N_{\epsilon}$  is number of times convergence is achieved before the algorithm stops. Finally, initial conditions were chosen stochastically by the algorithm itself, while the maximum number of functions evaluations, set to 1,000,000, was never achieved.

### A.2 Calibrating the covariance matrix scale factor

A key problem in implementing Metropolis algorithms is how to calibrate the covariance matrix's scale factor—the parameter c in subsection \_.\_\_in order to achieve an acceptance rate of the draws close to the ideal one (in high dimensions) of 0.23. Typically the problem is tackled by starting with some 'reasonable' value for c, and adjusting it after a certain number of iterations during the initial burn-in period. Specifically, given that the draws' acceptance rate is decreasing in c, c gets increased (decreased) if the initial acceptance rate was too high (low). A problem with this approach is that it does not guarantee that after the adjustment the acceptance rate will be reasonably close to the ideal one. The approach for calibrating c used in this paper, on the other hand—which is the same used in Benati (2008)—is based on the idea of estimating a reasonably good approximation to the inverse relationship between cand the acceptance rate by running a pre-burn-in sample. Specifically, let C be a grid of possible values for c—in what follows, we consider a grid over the interval [0.1, 1]with increments equal to 0.05. For each single value of c in the grid—call it  $c_i$ —we run n draws of the RWM algorithm as described in section 2.2.2, storing, for each  $c_j$ , the corresponding fraction of accepted draws,  $f_j$ . We then fit a third-order polynomial to the  $f_j$ 's via least squares, and letting  $\hat{a}_0$ ,  $\hat{a}_1$ ,  $\hat{a}_2$ , and  $\hat{a}_3$  be the estimated coefficients, we choose c by solving numerically the equation  $\hat{a}_0 + \hat{a}_1 c + \hat{a}_2 c^2 + \hat{a}_3 c^3 = 0.23$ . As the fraction of accepted draws reported in Table 1 shows, the procedure works quite well.

# B What Is the Additional State Variable Under Indeterminacy?

#### B.1 Reframing the issue in the correct way

Although intuitively sensible, from a strictly mathematical point of view the question 'What is the additional state variable under indeterminacy?' is unfortunately not well-posed, for the simple reason that state-space forms are unique up to a rotation. Specifically, the EMSSR

$$\xi_t^M = A\xi_{t-1}^M + Bu_t \tag{B.1}$$

$$Y_t = C\xi_t^M \tag{B.2}$$

—with  $Y_t$  still given by  $Y_t \equiv [R_t, \pi_t, y_t]'$  and  $C = [I_3 \ 0_{3 \times (N-3)}]$ , where N is the dimension of  $\xi_t^M$ —is, for a given 'input' vector  $u_t$ , observationally equivalent, in terms of 'output' vector  $Y_t$ , to the rotated state-space form

$$\tilde{\xi}_t^M = \tilde{A}\tilde{\xi}_{t-1} + \tilde{B}u_t \tag{B.3}$$

$$Y_t = \tilde{C}\tilde{\xi}_t \tag{B.4}$$

with  $\tilde{\xi}_t \equiv R\xi_t$ ,  $\tilde{A} \equiv RAR^{-1}$ ,  $\tilde{B} \equiv RB$ , and  $\tilde{C} \equiv CR^{-1}$ . In plain English, this means that for a *given* vector white noise input  $u_t$ , we can obtain exactly the same identical realisations for the observables *via* an infinity of state-space forms, all of which are a rotation of (B.1)-(B.2), and all of which are uniquely identified by a specific rotation matrix R.

So, from a mathematical point of view, asking oneself 'what' the states of a statespace form exactly 'are' is, strictly speaking, meaningless. The correct way of framing the issue is rather to focus directly on the VAR representation of the model. Let's start by defining as *minimal VAR representation* (henceforth, MVR) of the model a VAR representation for which the number of variables in the VAR is the same as the dimension of the state vector in the EMSSR minus the number of autoregressive disturbances—in the present case, three under determinacy and four under indeterminacy.<sup>21</sup> The crucial question then becomes:

'If, under indeterminacy, the first three variables in the MVR of the model are  $R_t$ ,  $\pi_t$ , and  $y_t$ , what is the fourth?'.

In order to answer this question, we first have to discuss a general mathematical property of this class of models.

 $<sup>^{21}</sup>$ The reason for focussing on the MVR is that (12), too, is a VAR representation of the model, but it suffers from the drawback of having an excess of variables.

## B.2 Exploring the set of admissible minimal VAR representations of the DSGE model

For obvious reasons, researchers always and uniquely focus upon VAR representations for DSGE models expressed in terms of observed variables—i.e., in the present case,  $R_t$ ,  $\pi_t$ , and  $y_t$ . This, however, overlooks a fact which becomes crucial in the present context: from a strictly mathematical point of view, the representation researchers typically fous upon is only one among an infinite number of admissible VAR representations. As we will illustrate shortly, in the present case this has the important implication that even if, under indeterminacy, we decide to arbitrarily restrict the first three variables in the MVR of the model to be  $R_t$ ,  $\pi_t$ , and  $y_t$ ,<sup>22</sup> still, the fourth variable is *entirely unconstrained*, in the specific sense that it can be *any* linear combination of the states in  $\xi_t^{23}$  (as we will elaborate below, we have no criteria to choose among them).

In order to illustrate this, let's start from the fact that—as it can be trivially shown—under both regimes it is possible to express the state vector of the original state-space form,  $\xi_t$ , as a function of the state vector of the EMSSR,  $\tilde{\xi}_t$ :

$$\xi_t = K \tilde{\xi}_t \tag{B.5}$$

where the dimension of K is  $N_0 \times N$ . Now, let H be an  $N \times N_0$  matrix subject to the only constraint that  $\tilde{K} \equiv H \cdot K$  be invertible, and let's define  $\tilde{Y}_t \equiv H\xi_t$ —in plain English,  $\tilde{Y}_t$  is an arbitrary linear combination of the states in the original statespace form. From (B.5) we therefore have that  $\tilde{Y}_t = \tilde{K}s_t$ , with  $\tilde{K}$  being  $N \times N$ . By substituting  $s_t = \tilde{K}^{-1}\tilde{Y}_t$  in (B.1), we therefore obtain  $\tilde{K}^{-1}\tilde{Y}_t = A\tilde{K}^{-1}\tilde{Y}_{t-1} + Bu_t$ , from which we get the set of all admissible MVRs of the model,  $\tilde{Y}_t = \tilde{K}A\tilde{K}^{-1}\tilde{Y}_{t-1} + \tilde{K}Bu_t$ , namely they

$$\tilde{Y}_t = \tilde{A}\tilde{Y}_{t-1} + \tilde{B}u_t \tag{B.6}$$

with  $\tilde{A}$  and  $\tilde{B}$  given by  $\tilde{K}A\tilde{K}^{-1}$  and  $\tilde{K}B$ , respectively.

Equation (B.6) expresses in mathematical form a fundamental principle: subject to the only constraint that the matrix H be such that  $\tilde{K}$  be invertible, any MVR for any linear combination of the states in the original state-space form is admissible to put it differently, the set of admissible MVRs of the DSGE models is of infinite dimension. Within the present context, this has the following implication. Even if, under indeterminacy, we decide to set the first three rows of H to  $[I_3 \ 0_{3\times 1}]$ —so that the first three elements of  $\tilde{Y}_t$  are  $R_t$ ,  $\pi_t$ , and  $y_t$ —still, the fourth element is entirely unconstrained, in the sense that it can be any linear combination of the states in  $\xi_t$ , subject to the only, previously mentioned invertibility constraint on  $\tilde{K}$ . The crucial

<sup>&</sup>lt;sup>22</sup>This restriction is intuitively appealing, as it obviously makes sense to focus on observed variables, rather than on arbitrary linear combinations of observed and unobserved variables (see below). Still, it is important to stress that, as we will show, it is, from a strictly mathematical point of view, entirely arbitrary.

 $<sup>^{23}</sup>$ Subject to the only trivial constraint that, in what follows, a specific matrix has to be invertible.

point to stress here is that, while we do have a compelling rationale for imposing that the first three elements of  $\tilde{Y}_t$  be  $R_t$ ,  $\pi_t$ , and  $y_t$ —they are the three observed variables—we have, on the other hand, *no* rationale at all for making a similar choice for the fourth element. Canova and Gambetti (2007) for example choose  $\pi_{t+1|t}$ , but by the very same logic we could choose  $y_{t+1|t}$ , or any linear combination of  $\pi_{t+1|t}$  and  $y_{t+1|t}$ , or, more generally,  $H_4.\xi_t$ , with  $H_4$  being a generic  $1 \times N$  vector.

Table 1	ayesian e	estimates of the	e struct	tural para	meters	
					Posterior distr	ibution: median
			Prior d	istribution	and 90%-cove	rage percentiles
				Standard	Before	After the
Parameter	Domain	Density	$\operatorname{Mode}$	deviation	October 1979	Volcker stabilisation
$\sigma^2_R$	+ 出	Inverse Gamma	0.25	0.25	0.492 [0.4	413; 0.592]
$\sigma^2_{\pi}$	+ 出	Inverse Gamma	0.5	0.5	0.391[0.5]	272; 0.562]
$\sigma^2_u$	+ 1	Inverse Gamma	0.1	0.25	0.055 [0.0	039; 0.078]
$\sigma_s^2$	+ 1	Inverse Gamma	0.25	0.25	0.193 [0.108; 0.406]	
X	+ Ľ	Gamma	0.05	0.01	0.044 [0.0	0.035; 0.056]
α	+ 1	$\operatorname{Gamma}$	2	1	8.062 [6.3	52; 10.434
α	[0, 1]	$\operatorname{Beta}$	0.75	0.2	0.059 [0.0	330; 0.099]
2	[0, 1]	$\operatorname{Beta}$	0.25	0.2	0.744 [0.0	388; 0.822]
θ	[0, 1)	$\operatorname{Beta}$	0.75	0.2	$0.595 \ [0.515; \ 0.670]$	$0.834 \ [0.779; \ 0.877]$
$\phi_{\pi}$	+ 1	Gamma	1.0	0.5	0.821 [0.701; 0.893]	$1.749 \ [1.107; \ 2.568]$
$\phi_y$	+	Gamma	0.15	0.25	0.527 [0.383; 0.672]	$1.146 \ [0.744; 1.610]$
$ ho_R$	[0, 1)	$\operatorname{Beta}$	0.25	0.2	0.404 [0.5]	289; 0.518]
$ ho_{\pi}$	[0, 1)	$\operatorname{Beta}$	0.25	0.2	0.418 [0.5	302; 0.521]
$ ho_y$	[0, 1)	$\operatorname{Beta}$	0.25	0.2	0.796 [0.7	738; 0.843]
Fraction of						
accepted draws					0.	246



Figure 1 Evolution of the coefficients of the  $VAR(\infty)$  representation of the model under indeterminacy, as a function of the lag order



Figure 2 Theoretical impulse-response functions under the two regimes





Figure 3 Replicating the Great Moderation: a typical stochastic simulation of the estimated DGP



Figure 4 Medians and 90%-coverage percentiles of the distributions of the estimated impulse-response functions to a 100 basis points increase in the nominal rate, under determinacy and indeterminacy (based on 1,000 replications)



#### **European Central Bank Working Paper Series**

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.europa.eu).

- 827 "How is real convergence driving nominal convergence in the new EU Member States?" by S. M. Lein-Rupprecht, M. A. León-Ledesma, and C. Nerlich, November 2007.
- 828 "Potential output growth in several industrialised countries: a comparison" by C. Cahn and A. Saint-Guilhem, November 2007.
- 829 "Modelling inflation in China: a regional perspective" by A. Mehrotra, T. Peltonen and A. Santos Rivera, November 2007.
- 830 "The term structure of euro area break-even inflation rates: the impact of seasonality" by J. Ejsing, J. A. García and T. Werner, November 2007.
- 831 "Hierarchical Markov normal mixture models with applications to financial asset returns" by J. Geweke and G. Amisano, November 2007.
- 832 "The yield curve and macroeconomic dynamics" by P. Hördahl, O. Tristani and D. Vestin, November 2007.
- 833 "Explaining and forecasting euro area exports: which competitiveness indicator performs best?" by M. Ca' Zorzi and B. Schnatz, November 2007.
- 834 "International frictions and optimal monetary policy cooperation: analytical solutions" by M. Darracq Pariès, November 2007.
- "435 "US shocks and global exchange rate configurations" by M. Fratzscher, November 2007.
- 836 "Reporting biases and survey results: evidence from European professional forecasters" by J. A. García and A. Manzanares, December 2007.
- 837 "Monetary policy and core inflation" by M. Lenza, December 2007.
- 838 "Securitisation and the bank lending channel" by Y. Altunbas, L. Gambacorta and D. Marqués, December 2007.
- 839 "Are there oil currencies? The real exchange rate of oil exporting countries" by M. M. Habib and M. Manolova Kalamova, December 2007.
- 840 "Downward wage rigidity for different workers and firms: an evaluation for Belgium using the IWFP procedure" by P. Du Caju, C. Fuss and L. Wintr, December 2007.
- 841 "Should we take inside money seriously?" by L. Stracca, December 2007.
- 842 "Saving behaviour and global imbalances: the role of emerging market economies" by G. Ferrucci and C. Miralles, December 2007.
- 843 "Fiscal forecasting: lessons from the literature and challenges" by T. Leal, J. J. Pérez, M. Tujula and J.-P. Vidal, December 2007.
- 844 "Business cycle synchronization and insurance mechanisms in the EU" by A. Afonso and D. Furceri, December 2007.
- 845 "Run-prone banking and asset markets" by M. Hoerova, December 2007.

- 846 "Information combination and forecast (st)ability. Evidence from vintages of time-series data" by C. Altavilla and M. Ciccarelli, December 2007.
- 847 "Deeper, wider and more competitive? Monetary integration, Eastern enlargement and competitiveness in the European Union" by G. Ottaviano, D. Taglioni and F. di Mauro, December 2007.
- 848 "Economic growth and budgetary components: a panel assessment for the EU" by A. Afonso and J. González Alegre, January 2008.
- 849 "Government size, composition, volatility and economic growth" by A. Afonso and D. Furceri, January 2008.
- 850 "Statistical tests and estimators of the rank of a matrix and their applications in econometric modelling" by G. Camba-Méndez and G. Kapetanios, January 2008.
- 851 "Investigating inflation persistence across monetary regimes" by L. Benati, January 2008.
- 852 "Determinants of economic growth: will data tell?" by A. Ciccone and M. Jarocinski, January 2008.
- 853 "The cyclical behavior of equilibrium unemployment and vacancies revisited" by M. Hagedorn and I. Manovskii, January 2008.
- 854 "How do firms adjust their wage bill in Belgium? A decomposition along the intensive and extensive margins" by C. Fuss, January 2008.
- 855 "Assessing the factors behind oil price changes" by S. Dées, A. Gasteuil, R. K. Kaufmann and M. Mann, January 2008.
- 856 "Markups in the euro area and the US over the period 1981-2004: a comparison of 50 sectors" by R. Christopoulou and P. Vermeulen, January 2008.
- 857 "Housing and equity wealth effects of Italian households" by C. Grant and T. Peltonen, January 2008.
- 858 "International transmission and monetary policy cooperation" by G. Coenen, G. Lombardo, F. Smets and R. Straub, January 2008.
- 859 "Assessing the compensation for volatility risk implicit in interest rate derivatives" by F. Fornari, January 2008.
- 860 "Oil shocks and endogenous markups: results from an estimated euro area DSGE model" by M. Sánchez, January 2008.
- 861 "Income distribution determinants and public spending efficiency" by A. Afonso, L. Schuknecht and V. Tanzi, January 2008.
- 862 "Stock market volatility and learning" by K. Adam, A. Marcet and J. P. Nicolini, February 2008.
- 863 "Population ageing and public pension reforms in a small open economy" by C. Nickel, P. Rother and A. Theophilopoulou, February 2008.
- 864 "Macroeconomic rates of return of public and private investment: crowding-in and crowding-out effects" by A. Afonso and M. St. Aubyn, February 2008.
- 865 "Explaining the Great Moderation: it is not the shocks" by D. Giannone, M. Lenza and L. Reichlin, February 2008.
- 866 "VAR analysis and the Great Moderation" by L. Benati and P. Surico, February 2008.

